Author's response to reviews

Title: Evaluating the systematic implementation of the 'Let Me Decide' advance care planning programme in long term care through focus groups: Staff perspectives.

Authors:

Nicola Cornally (n.cornally@ucc.ie)
Ciara McGlade (ciaramcglade@gmail.com)
Elizabeth Weathers (e.weathers@ucc.ie)
Edel Daly (edel.daly66@gmail.com)
Carol Fitzgerald (carol.fitzgerald@ucc.ie)
Rónán O’Caoimh (rocaoimh@hotmail.com)
Alice Coffey (a.coffey@ucc.ie)
William D Molloy (w.molloy@ucc.ie)

Version: 4 Date: 2 July 2015

Author's response to reviews:

Cover letter attached containing response to editors comments.